Author's response to reviews

Title: Calibrating ADL-IADL scales to improve measurement accuracy and to extend the disability construct into the preclinical range: a systematic review

Authors:

Robert A Fieo (r.fieo@sms.ed.ac.uk)
Elizabeth J Austin (elizabeth.austin@ed.ac.uk)
John M Starr (jstarr@staffmail.ed.ac.uk)
Ian J Deary (iand@staffmail.ed.ac.uk)

Version: 8 Date: 8 August 2011

Author's response to reviews:

typographical changes